Author's response to reviews

Title: Incidence and cost of Treatment-Emergent Comorbid Events in Insured Patients with Chronic Hepatitis C Virus Infection: A Retrospective Cohort Study

Authors:

Sandhya Sapra (sandhya.sapra@bms.com)
Eunice Chang (echang@pharllc.com)
Michael S Broder (mbroder@pharllc.com)
Gilbert L'Italien (gilbert.litalien@bms.com)

Version: 2
Date: 17 March 2014

Author's response to reviews:

March 17, 2014

Christopher Morrey, PhD
Executive Editor
BMC Health Services Research

Dear Dr. Morrey:

I have made the following changes requested by BMC Health Services Research.

1. An ethics statement has been included in the Methods section of the text, declaring that an approval by the ethics committee was not needed as no identifiable information was used to conduct the study.
2. Line numbering was added to the manuscript.
3. The Acknowledgements section was added to the text.

Thank you for your time and consideration. We look forward to your decision.

Sincerely,

Michael S. Broder, MD, MSHS
Partnership for Health Analytic Research, LLC
280 S. Beverly Dr., Suite 404, Beverly Hills, CA 90212
Phone: 310-858-9555; Fax: 310-858-9552; mbroder@pharllc.com